\newpage
\addcontentsline{toc}{chapter}{List of publications}

\begingroup
%\continuouslabelsfalse
\labelsep 0pt
\makeatletter
\def\@biblabel#1{\relax}
\makeatother
\bibliographypub{../../bibdb/bibdb}{}
\bibliographystylepub{thesispub}
\endgroup

% own publications
\nocitepub{zirkelbach11,zirkelbach10,zirkelbach09}
\nocitepub{zirkelbach2007,zirkelbach2006,zirkelbach2005}

%\renewcommand\bibname{List of publications}

